Author's response to reviews

Title: Myocardial Contractility in the echo lab: molecular, cellular and pathophysiological basis.

Authors:

Tonino Bombardini (tbombardini@yahoo.it)

Version: 2 Date: 26 August 2005

Author's response to reviews:

Unnecessary capitalisation has been removed
The affiliation has been corrected
The acknowledgements are moved to after the conclusion
The tables are moved to after the reference list
The figure legends are moved to after the tables